Author’s response to reviews

Title: Nal-IRI with 5-fluorouracil (5-FU) and leucovorin or gemcitabine plus cisplatin in advanced biliary tract cancer - The NIFE trial (AIO-YMO HEP-0315) an open label, non-comparative, randomized, multicenter phase II study

Authors:

Lukas Perkhofer (lukas.perkhofer@uniklinik-ulm.de)
Andreas Berger (andreas.berger@vivantes.de)
Alica Beutel (alica.beutel@uniklinik-ulm.de)
Eike Gallmeier (eike.gallmeier@med.uni-marburg.de)
Stefan Angermeier (stefan.angermeier@kliniken-lb.de)
Ludwig Fischer von Weikersthal (weikersthal.ludwig@gesundheitszentrum.klinikum-amberg.de)
Thorsten Goetze (Goetze.Torsten@khnw.de)
Rainer Muche (rainer.muche@uni-ulm.de)
Thomas Seufferlein (thomas.seufferlein@uniklinik-ulm.de)
Thomas Ettrich (thomas.ettrich@uniklinik-ulm.de)

Version: 2 Date: 20 Aug 2019

Author’s response to reviews:

Dear Editorial Office,

attached you can find the minor changes made accordingly.

1. Please ensure that the corresponding author listed in the manuscript matches the corresponding author listed in the submission system - currently they differ.
   - the corresponding author is changed in the Submission System

2. In the 'Trial Registration' section of the abstract, please include the date of registration.
   - the registration date is included in the Manuscript

3. Thank you for providing a SPIRIT checklist. As this document is no longer required at this stage of
the publication process, please remove your SPIRIT checklist from your submission’s supplementary files.
- the SPIRIT diagram is removed

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
- accordingly uploaded.

Kind regards,

Lukas Perkhofer